Author’s response to reviews

Title: The Effect on Knee Joint Load of Instruction in Analgesic Use Compared With Neuromuscular Exercise in Patients With Knee Osteoarthritis: Protocol for a randomized, single-blind, controlled trial (The EXERPHARMA trial)

Authors:

Brian Clausen (brclausen@health.sdu.dk)
Anders Holsgaard-Larsen (ahlarsen@health.sdu.dk)
Jens Søndergaard (jsoendergaard@health.sdu.dk)
Robin Christensen (Robin.Christensen@frh.regionh.dk)
Thomas P Andriacchi (tandriac@stanford.edu)
Ewa M Roos (eroos@health.sdu.dk)

Version: 2 Date: 23 June 2014

Author’s response to reviews:

Thank you for going through the manuscript.

I have uploaded a new version of the manuscript and updated a figure.

Date of trial registration has been added to page 5.

Informed consent was obtained as described on page 10 line 6-8.

Legends for the figures have been provided at page 33. Figure 2 has been amended to fit the legend.

You have asked for documentation of ethical approval and funding. This will be provided by separate e-mail. Please note that the original documentation is in Danish, translation has been provided as well (done by Google Translate).

Kind regards

Brian Clausen